Author's response to reviews

Title: Evaluation of two formulations of adjuvanted RTS,S malaria vaccine in children aged 3 to 5 years living in a malaria-endemic region of Mozambique: a Phase I/II randomized double-blind bridging trial

Authors:

Eusebio EM Macete (eusebiomacete@yahoo.com)
Jahit JS Sacarlal (jahit.sacarlal@manhica.net)
John JA Aponte (jjairo@clinic.ub.es)
Amanda AL Leach (amanda.leach@gskbio.com)
Margarita MN Navia (mnavia@pcb.ub.es)
Jessica JM Milman (jmilman@malariaivaccine.org)
Caterina CG Guinovart (cguinova@clinic.ub.es)
Inacio IM Mandomando (inacio.mandomando@manhica.net)
Mateu ME Espasa (mespasa2002@yahoo.com)
Yolanda YL Lopez -Pua (YOLOPEZ@clinic.ub.es)
Marc ML Lievens (marc.lievens@gskbio.com)
Alex AO Owusu-Ofori (alex.owusu@gskbio.com)
Marie-Claude MCD Dubois (marie-claude.dubois@gskbio.com)
Conor CC Cahill (conor.cahill@gskbio.com)
Marguerite MK Koutsoukos (Marguerite.Koutsoukos@gskbio.com)
Marla MS Sillman (msillman@malariaivaccine.org)
Ricardo RT Thompson (rthompson@tvcabo.co.mz)
Filip FD Dubovsky (fdubovsky@malariaivaccine.org)
W Ripley RB Ballou (ripley.ballou@gskbio.com)
Joe JC Cohen (joe.cohen@gskbio.com)
Pedro L Alonso (palonso@clinic.ub.es)

Version: 4 Date: 31 August 2006

Author’s response to reviews: see over
Dear editor,

Please find enclosed our manuscript entitled:

"Evaluation of two formulations of adjuvanted RTS,S malaria vaccine in children aged 3 to 5 years living in a malaria-endemic region of Mozambique: a Phase I/II randomised double-blind bridging trial", which we would like to be considered for publication in Trial.

The data presented here are original, have not been published previously and are not currently under revision elsewhere.

This version is an update of the previous one that was sent to Trial. I have made some changes such as adding the trial registration number and some other small changes, such as removing the figures (as they are uploaded separately).

We thank you in advance for your kind attention.

Yours sincerely,

Eusebio Macete

Eusebio Macete, MD, MPH
eusebiomacete@yahoo.com

Centre de Salut Internacional, Hospital Clínic/IDIBAPS, Calle Roseelló 132 2º, 2ª, Universitat de Barcelona, Barcelona - 08036, Spain
and
Centro de Investigação em Saúde da Manhiça (Manhiça Health Research Centre)
C.P. Nº 1929 Maputo - Mozambique
Mozambique